Author's response to reviews

Title: Internet-based Self-Assessment after the Tsunami: lessons learned

Authors:

Stefan Vetter (stefan.vetter@access.uzh.ch)
Astrid Rossegger (astrid.rossegger@me.com)
Thomas Elbert (thomas.elbert@uni-konstanz.de)
Juliane Gerth (juliane.gerth@gmail.com)
Frank Urbaniok (frank.urbaniok@ji.zh.ch)
Arja Laubacher (arja.laubacher@ji.zh.ch)
Wulf Rossler (roessler@dgsp.uzh.ch)
Jérôme Endrass (jerome.endrass@me.com)

Version: 11 Date: 5 January 2011

Author's response to reviews:

One address of an author has changed in the mean time and was adapted in the manuscript as final changes!